This article is intended solely for the personal use of the individual user and is not to be disseminated broadly

This document is copyrighted by the American Psychological Association or one of its allied publishers.

MERICAN

https://doi.org/10.1037/pag0000749

Initial Status and Change in Cognitive Function Mediate the Association Between Academic Education and Physical Activity in Adults Over 50 Years of Age

Boris Cheval^{1, 2}, Ilyes Saoudi³, Silvio Maltagliati³, Layan Fessler³, Ata Farajzadeh^{4, 5},

Stefan Sieber⁶, Stéphane Cullati^{7, 8}, and Matthieu P. Boisgontier^{4, 5}

¹ Swiss Center for Affective Sciences, University of Geneva

² Department of Psychology, Laboratory for the Study of Emotion Elicitation and Expression (E3Lab), University of Geneva

³ Laboratoire Sport et Environnement Social (SENS), Unité de Formation et de Recherche en Sciences et Techniques des Activités

Physiques et Sportives (UFR STAPS), Université Grenoble Alpes

⁴ School of Rehabilitation Sciences, Faculty of Health Sciences, University of Ottawa

⁵ Bruyère Research Institute, Élisabeth Bruyère Hospital, Ottawa, Canada

⁶ LIVES Centre, Swiss Centre of Expertise in Life Course Research, University of Lausanne

⁷ Population Health Laboratory, University of Fribourg

⁸ Department of Readaptation and Geriatrics, University of Geneva

Higher levels of academic education are associated with higher levels of physical activity throughout the lifespan. However, the mechanisms underlying this association are unclear. Cognitive functioning is a potential mediator of this association because higher levels of education are associated with better cognitive function, which is related to greater engagement in physical activity. Here, we used large-scale longitudinal data from 105,939 adults 50 years of age or older (55% women) from the Survey of Health, Ageing, and Retirement in Europe to investigate whether initial status and change in cognitive function mediate the relationship between education and change in physical activity. Education and physical activity were selfreported. Cognitive function was assessed based on delayed recall and verbal fluency. Academic education was assessed at the first measurement occasion. The other measures were collected seven times between 2004 and 2019. The mediating role of cognitive function was tested using longitudinal mediation analyses combined with growth curve models. We found that higher levels of education were associated with higher levels and slower decreases in cognitive function, which in turn predicted a lower decrease in physical activity across time. These results support the presence of an indirect effect of education on physical activity trajectories by affecting the intercept and slope of cognitive function. Specifically, these findings suggest that both the initial status and change in cognitive function mediate the association between academic education and change in physical activity. In addition, results revealed that, across the aging process, differences in cognitive function and physical activity widen between the low and high educated. In other words, this study demonstrates the long-lasting effect of education on cognitive function and physical activity.

Public Significance Statement

The beneficial effects of education on cognition and physical activity are well-replicated. Yet, little is known about the mechanisms underlying these associations. Based on a large-scale longitudinal European study in adults aged 50 years or older, we show that both the initial status of cognitive function and its change underpin the relationship between education and physical activity across time. Our results shed light on education that engenders positive trajectories of cognition that in turn impact physical activity. These findings highlight the long-run effect of education on cognitive function and physical activity.

Keywords: educational status, cognition, exercise, aging, longitudinal studies

Supplemental materials: https://doi.org/10.1037/pag0000749.supp

Boris Cheval D https://orcid.org/0000-0002-6236-4673

This article uses data from Survey of Health, Ageing, and Retirement in Europe (SHARE) Waves 1, 2, 4, 5, 6, 7, and 8 (https://doi.org/10.6103/ SHARE.w1.800, https://doi.org/10.6103/SHARE.w2.800, https://doi.org/ 10.6103/SHARE.w4.800, https://doi.org/10.6103/SHARE.w5.800, https:// doi.org/10.6103/SHARE.w6.800, https://doi.org/10.6103/SHARE.w7.800, https://doi.org/10.6103/SHARE.w8ca.800). The SHARE data collection was primarily funded by the European Commission through FP5 (QLK6CT-2001-00360), FP6 (SHARE-I3: RII-CT-2006-062193, COMPARE: CIT5-CT-2005-028857), and FP7 (SHARE-PREP: No. 211909, SHARE-LEAP: No. 227822, SHARE M4: No. 261982). Additional funding from the German Ministry of Education and Research, the Max Planck Society for the Advancement of Science, the U.S. National Institute on Aging (U01_AG09740-13S2, P01_AG005842, P01_AG08291, P30_AG12815, R21_AG025169, Y1-AG-4553-01, IAG_BSR06-11, OGHA_04-064, HHSN271201300071C), and various national funding sources is gratefully acknowledged (see https:// www.share-project.org).

Academic educational attainment plays a crucial role in promoting the engagement in multiple health behaviors (Conti et al., 2010; Eide & Showalter, 2011; Silles, 2009; Zajacova & Lawrence, 2018), such as physical activity (Chalabaev et al., 2022; Cheval, Sieber, et al., 2018; Droomers et al., 2001; Kari et al., 2020). Less educated people exhibit a higher risk of physical inactivity, a risk that increases with age (Beenackers et al., 2012; Cheval, Sieber, et al., 2018; Droomers et al., 2001; O'Donoghue et al., 2018). Moreover, a recent Mendelian randomization study conducted on middle-age Finnish adults revealed that a higher level of education was associated with more time spent on physical activity (Kari et al., 2020). Overall, these findings support the role of education in determining physical activity levels and their maintenance across the life course. Still, gaps remain in our understanding of the mechanisms underpinning this association.

Several mechanisms may explain why lower levels of education have been associated with lower physical activity. First, poor health literacy has been suggested to explain the relationship between lower education and lower engagement in protective health behaviors (Friis et al., 2016; Levin-Zamir et al., 2016; Stormacq et al., 2019; Sun et al., 2013; Van Der Heide et al., 2013). Second, less educated people may have developed weaker motivation and self-efficacy toward healthprotective behaviors, putting them at higher risk for disengagement from these behaviors (Nettle, 2010; Osborn et al., 2011; Torres & Marks, 2009; Verhoeven & Snow, 2001). Third, because the place of residence is influenced by sociocultural position (Diez Roux & Mair, 2010), less educated people are more likely to live in disadvantaged physical (e.g., poor-maintained facilities, lack of aesthetic, and natural spaces) and social environments (e.g., unsafe and isolated areas), which could contribute to their lack of engagement in physical activity (Cheval et al., 2019; Rees-Punia et al., 2018; Xiao et al., 2018).

In addition to these mechanisms, cognitive function is a potential candidate to mediate the association between education and physical activity. On the one hand, previous studies and meta-analyses showed that higher levels of education were associated with better cognitive function (Aartsen et al., 2019; Anstey & Christensen, 2000; Avila et al., 2009; Lenehan et al., 2015; Opdebeeck et al., 2016; Schneeweis et al., 2014). On the other hand, studies showed that better cognitive function predicted higher engagement in physical activity (Cheval, Boisgontier, et al., 2022; Cheval, Csajbók, et al., 2021; Cheval, Orsholits, et al., 2020; Daly et al., 2015; Sabia et al., 2017). According to the theory of effort minimization in physical activity (TEMPA; Cheval & Boisgontier, 2021), this latter association is explained by the fact that cognitive function is considered necessary to counteract the innate human attraction to effort minimization (Cheval & Boisgontier, 2021; Cheval et al., 2017; Cheval, Radel, et al., 2018; Klein-Flügge et al., 2016; Lieberman, 2015; Prévost et al., 2010). In line with this view and consistent with epidemiological and neurobehavioral evidence,

experimental studies have demonstrated that cognitive function is crucial in inhibiting this automatic attraction to physical inactivity and in promoting regular engagement in physical activity (Cheval, Cabral, et al., 2021; Cheval, Daou, et al., 2020; Cheval, Tipura, et al., 2018). In sum, existing literature suggests that cognitive function could underlie the association between education and physical activity. However, to the best of our knowledge, no study has directly investigated this potential mediation.

In addition, studies showed that physical activity enhances cognitive function (Blondell et al., 2014; Cheval, Darrous, et al., 2023; Hamer et al., 2018; Hillman et al., 2008), suggesting a reciprocal relationship between these variables (Cheval, Orsholits, et al., 2020; Csajbók et al., 2022). However, recent studies based on the Survey of Health, Ageing, and Retirement in Europe (SHARE) data set formally tested these bidirectional associations using bivariant latent change score models (Cheval, Orsholits, et al., 2020) and longitudinal mediation models combined with autoregressive cross-lagged models (Csajbók et al., 2022). Results showed that changes in cognitive function mainly preceded changes in physical activity. Yet, the effect from physical activity to cognitive function was also significant, albeit weaker. These results suggest a detrimental circular effect between cognitive decline and physical inactivity, with lower cognitive function increasing physical inactivity, which in turn further increase cognitive decline. Here, we focused on the potential mediating role of cognitive function in the relationship between academic education and physical activity (i.e., education \rightarrow cognitive function \rightarrow physical activity). To provide a finer grained understanding of the mechanisms underlying this potential mediation, two aspects of cognitive function were considered (initial status and change). This approach allowed disentangling the mediation effects of baseline level and growth in the cognitive function.

The objective of our study was to examine whether initial status and change in cognitive function mediate the relationship between education and change in physical activity in adults 50 years of age or older. Based on previous literature, we hypothesized that higher levels and slower decline in cognitive function mediate the relationship between academic education and physical activity over time.

Method

Transparency and Openness

The SHARE data set used for this study is publicly available at http://www.share-project.org/data-access.html (Börsch-Supan, 2022a, 2022b, 2022c, 2022d, 2022e, 2022f, 2022g). The code of the analyses is available at https://zenodo.org/deposit/7520036. Please see Author note section for details. The hypotheses and analyses were not

Correspondence concerning this article should be addressed to Boris Cheval, Swiss Center for Affective Sciences, University of Geneva, Campus Biotech, Chemin des Mines 9, 1202 Genève, Switzerland. Email: boris.cheval@unige.ch

Boris Cheval is supported by an Ambizione Grant (PZ00P1_180040) from the Swiss National Science Foundation. Matthieu P. Boisgontier is supported by the Natural Sciences and Engineering Research Council of Canada (RGPIN-2021-03153) and a grant from The Banting Research Foundation. All participants provided written informed consent.

Boris Cheval contributed in conceptualization and writing-original draft. Ilyes Saoudi contributed in writing-review and editing. Silvio Maltagliati contributed in writing-review and editing. Layan Fessler contributed in writing-review and editing. Ata Farajzadeh contributed in writing-review and editing. Stefan Sieber contributed in data curation and writing-review and editing. Stefan Cullati contributed in writing-review and editing.

Matthieu P. Boisgontier contributed in supervision and writing-review and editing.

An earlier version of this article was preprinted at https://doi.org/ 10.51224/SRXIV.151 (Cheval, 2023). The SHARE data set is available at http://www.share-project.org/data-access.html. The scripts for data management (i.e., from raw data to data used in the analyses) and for the models tested are available at https://zenodo.org/deposit/7520036.

preregistered but are part of a larger project aiming to understand the life course influences on health trajectories at older age.

Participants and Study Design

Data were drawn from SHARE, a longitudinal population-based study of adults 50 years of age or older living in 27 European countries and one middle east country (Börsch-Supan et al., 2013). Data were collected every 2 years between 2004 and 2019, with a total of eight measurement waves using computer-assisted personal interviews in participants' homes. Physical activity and cognitive function (delayed recall and verbal fluency) were assessed in all measurement waves except Wave 3 (2008-2009). Academic education level was measured when participants were first included in the study. The SHARE study was approved by the relevant research ethics committees in the participating countries, and all participants provided written informed consent. To be included in the analyses, at least one measure of physical activity, one measure of cognitive function, and indicated their education level was required. We excluded individuals with suspected dementia at baseline (N = 3,089), as indicated by a score above two on the time-orientation question (Aartsen et al., 2019), and individuals who reported more than two limitations in activities of daily living at baseline (N = 10,546).

Measures

Outcome: Physical Activity

Physical activity was derived from the following question: "How often do you engage in activities that require a low or moderate level of energy such as gardening, cleaning the car, or going for a walk?" (Boisgontier et al., 2020; Cheval et al., 2019; Cheval, Orsholits, et al., 2020; de Souto Barreto et al., 2017). Participants responded using a 4-point scale: $1 = more \ than \ once \ a \ week; 2 = once \ a \ week; 3 = one \ to \ three \ times \ a \ month; 4 = hardly \ ever \ or \ never$. In the models, this measure was treated as continuous and ranged from 0 to 3.

Predictor Variable: Education

The level of education was based on the United Nations Educational, Scientific and Cultural Organization's International Standard Classification of Education (ISCED; United Nations Educational, 2006) differentiating primary (ISCED Levels 0 and 1), secondary (ISCED Levels 2 to 4), and tertiary level of education (ISCED Levels 5 and 6; Cullati et al., 2018). In the models, this measure was treated as continuous and ranged from 0 to 2.

Mediating Variables: Cognitive Function

Cognitive function was assessed using two indicators including delayed recall and verbal fluency. Delayed recall is a reliable predictive measure of the development of dementia (Sano et al., 2011; Zhao et al., 2012) and was extracted from an adapted 10-word delayed recall test (Harris & Dowson, 1982). First, participants listened to a list of 10 words that were read out loud by the interviewer. Then, they were immediately asked to recall as many words as possible. At the end of the cognitive testing session,

the participants were asked to recall the words from the list a second time, which captured a delayed recall score. This score ranged from 0 to 10, with higher scores indicating a better cognitive function. In the verbal fluency test (Rosen, 1980), the interviewer asked participants to name as many different animals as they could think of in 1 min. The score was the total number of correctly named animals, with a higher score indicating better performance.

Covariates

All the models were adjusted for age group (50–64, 65–79, 80–96 years) and sex (male, female). Note that health-related variables (e.g., depressive symptoms, self-rated health) were not included because they could reasonably be considered colliders based on causal graphs.

Statistical Analysis

First, growth curve models were used to estimate the evolution of the indicators of cognitive function and physical activity across the waves of measurement. These models were adjusted for age group (50-64, 65-79, 80-96 years), sex (male, female), and the level of education. Then, the mediating role of cognitive function on the association between education and physical activity was analyzed using longitudinal mediation analyses combined with growth curve models (von Soest & Hagtvet, 2011). The mediating paths were defined as follows: Academic education (time-invariant exogenous variable) predicted the initial status (intercept) of cognitive function (Path a1), which predicted the change (slope) in physical activity (Path b1). In addition, academic education predicted the change in cognitive function (Path a2), which predicted the change in physical activity (Path b2). A direct effect from education to change in physical activity was included (c'). Since the initial status of physical activity was likely to be correlated with its subsequent change in physical activity, a path from initial status of physical activity to change in physical activity was included. This path allowed estimating differences in physical activity changes as a function of education while adjusting for differences in initial status (Seltzer et al., 2003; von Soest & Hagtvet, 2011). In other words, this path allowed us to obtain a mediation analysis that was not confounded by differences in the initial status of the dependent variable (Littlefield et al., 2010). The equations of the models using the terminology of von Soest and Hagtvet (2011) are as follows:

$$\gamma_{it} = \alpha_{iv} + \beta_{iv}\lambda_t + \varepsilon_{vit}, \tag{1}$$

$$\alpha_{i\omega} = \mu_{\alpha\omega} + y_{\alpha\omega}\chi_i + \zeta_{\alpha\omega i}, \qquad (2)$$

$$\beta_{i\omega} = \mu_{\beta\omega} + y_{\beta\omega}\chi_i + \zeta_{\beta\omega i}, \qquad (3)$$

$$\beta_{iy} = \mu_{\beta y} + \delta_{\alpha y} \alpha_{iy} + \delta_{\alpha \omega} \alpha_{i\omega} + \delta_{\beta \omega} \beta_{i\omega} + \gamma_{\beta y'} \chi_i + \zeta_{\beta yi}.$$
(4)

In Equation 1, γ identifies all parameters predicting, γ_{it} *i* represents the individual, and *t* represents the time points. Thus, γ_{it} is the value of the trajectory of the outcome variable (i.e., physical activity) for individual *i* at Time *t*. α_{iy} is the random intercept for every individual *i*. β_{iy} is the random slope for every individual *i*. λ_t is a parameter used to identify time. ε_{yit} is the error term at each time point for each individual.

In Equation 2, $\mu_{\alpha\varepsilon}$ is the intercept of the equation that predicts the random intercept $\alpha_{i\omega}$ of the mediating variable (i.e., cognitive function), $y_{\alpha\omega}$ is the coefficient of regression of the predictor variable (i.e., education), $\zeta_{\alpha\omega}$ is the error term, χ_i is the predictor variable (i.e., education), and *i* identifies individuals' data points.

In Equation 3, $\mu_{\beta\omega}$ is the intercept of the equation that predict the random slope of the mediating variable $\beta_{i\omega}$, $y_{\beta\omega}$ is the regression coefficient of the predictor variable, $\zeta_{\beta\omega i}$ is the error term.

In Equation 4, $\mu_{\beta y}$ is the intercept of the equation that predicts the random slope β_{iy} of the dependent variable (i.e., physical activity). The slope of the dependent variable is predicted by the predictor variable (i.e., education) χ_i and the random intercept of the mediating variable $\alpha_{i\omega}$. $\gamma_{\beta y'}$ is the regression coefficient for the predictor variable. $\delta \alpha \omega$ is the regression coefficient for the intercept of the mediating variable. $\delta_{\beta \omega}$ is the regression coefficient from the slope of the mediating variable to the slope of the dependent variable. $\delta_{\alpha y}$ is the coefficient for the regression of the slope of the dependent variable on its own intercept α_{iy} . $\zeta_{\beta yi}$ represents the error term. Thus, in Model 3, $\gamma_{\beta y'}$ represents the expected difference in physical activity growth rates as a function of the level of education and adjusted for differences in initial physical activity levels. The longitudinal mediation model is illustrated in Figure 1.

Indirect effects at the intercept levels (indirect $L = y_{\alpha\omega} \times \delta_{\alpha\omega}$) and slope levels (indirect $S = y_{\beta\omega} \times \delta_{\beta\omega}$), as well as the total effect ($\gamma_{\beta\gamma} = \gamma_{\beta\gamma'} + y_{\alpha\omega} \times \delta_{\alpha\omega} + y_{\beta\omega} \times \delta_{\beta\omega}$) were estimated using maximum likelihood estimator and full information maximum likelihood estimation for the missing values (Maydeu-Olivares, 2017). The proportions of the total effects explained by the mediating variable at the intercept and slope levels were calculated by subtracting the direct effect $(\gamma_{\beta\gamma'})$ from the total effect $(\gamma_{\beta\gamma})$, and then dividing the result by the total effect (i.e., [total effect –direct effect]/total effect). The analyses were performed with the lavaan R package (Rosseel, 2012). The percentage of variance explained in the slope of physical activity were computed in lavaan.

Sensitivity Analyses

Two sensitivity analyses were conducted. In the first sensitivity analysis, two additional paths were specified in the model (i.e., from education to the intercept of physical activity and from the intercept of cognitive function to the slope of cognitive function). These paths were added to check that the main findings were not biased by the absence of these direct paths. Second, to satisfy the temporal precedence between the slope of cognitive function and the slope of physical activity, we created a time lag between cognitive function (i.e., the mediator) and physical activity (i.e., the outcome). Accordingly, the slope of the cognitive function was estimated before the slope of physical activity, thereby satisfying the causal condition of temporal precedence between the mediator and the dependent variable (Cheong et al., 2003; MacKinnon, 2012).

Results

The scripts for data management (i.e., from raw data to data used in the analyses) and for the models tested are available at https://zeno do.org/deposit/7520036. In addition, the complete results of the models are included in the Supplemental Material.





Note. Education is the predictor variable. Cognitive function (i.e., delayed recall or verbal fluency) is the mediating variable. Physical activity is the dependent variable. This model tests whether the intercept and the slope of cognitive function mediate the association between education and change in physical activity.

Descriptive Results

A total of 105,939 participants (55% women) were included in the sample. Table 1 presents the characteristics of the participants stratified by education level. Simple association tests showed that participants with higher levels of education had better cognitive function, higher levels of physical activity, were younger, and more likely to be a man (p < .001). Moreover, during follow-up, 10,043 (9.5%) participants died, 35,179 (33.2%) dropped out (i.e., did not respond to Wave 7 or 8), and 60,717 (57.3%) continued to participate in the study. On average, participants had 2–3 repeated measures of cognitive function and physical activity. More specifically, 37,434, 25,293, 19,032, 12,417, 4,678, 4,213, and 2,872, responded to one, two, three, four, five, six, and seven waves of measurement, respectively.

Growth Curves Models

Physical Activity

Results of the growth curves model adjusted for age group (ref. 50–64 years), gender (ref. women), and the level of education (ref. primary) showed a significant latent intercept (b = 2.459, 95% confidence interval [95% CI 2.442, 2.476], p < .001) and a significant negative latent slope (b = -0.030, 95% CI [-0.034, 0.026], p < .001), indicating a decrease in the level of physical activity over time. The negative latent slope was more than twice higher in adults 65–79 years of age (b = -0.069, 95% CI [-0.073, -0.065], p < .001) and was more than thrice higher in adults 80–96 years of age (b = -0.100, 95% CI [-0.109, 0.091], p < .001).

Delayed Recall

Results of the growth curves model adjusted for age group (ref. 50–64 years), gender (ref. women), and the level of education (ref. primary) showed a significant latent intercept (b = 3.491, 95% CI [3.458, 3.525], p < .001) and a significant positive latent slope (b = 0.011, 95% CI [0.004, 0.0019], p = .002), indicating an increase in the level of delayed recall across time in adults 50–64 years of age. However, a significant negative latent slope was observed in adults

Table 1

Baseline Characteristics of the Participants Across Levels of Education

65–79 (b = -0.069, 95% CI [-0.076, -0.061], p < .001) and 80–96 years of age (b = -0.086, 95% CI [-0.104, -0.068], p < .001).

Verbal Fluency

Results of the growth curves model adjusted for age group (ref. 50–64 years), gender (ref. women), and the level of education (ref. primary) showed a significant latent intercept (b = 1.727, 95% CI [1.715, 1.738], p < .001), but a nonsignificant latent slope (b = 0.001, 95% CI [-0.001, 0.003], p < .001), indicating no significant change in the level of verbal fluency across time in adults 50–64 years of age. However, a significant negative latent slope was observed in adults 65–79 (b = -0.029, 95% CI [-0.031, -0.026], p < .001) and 80–96 years of age (b = -0.043, 95% CI [-0.049, -0.037], p < .001).

Longitudinal Mediation Analyses

Table 2 and Figure 2 show the results of the longitudinal mediation analyses combined with growth curve models.

Delayed Recall

We observed a significant association of education with the intercept of delayed recall ($y_{\alpha\omega} = 0.809, 95\%$ CI [0.786, 0.832], p < .001) and a significant association of the intercept of delayed recall with the slope of physical activity ($\delta_{\alpha\omega} = 0.010, 95\%$ CI [0.008, 0.012], p < .001). Thus, the intercept of delayed recall mediated the association between education and the slope in physical activity, as confirmed by a significant indirect effect at this intercept level (indirect effect L = 0.008, 95% CI [0.007, 0.010], p < .001). In addition, results showed a significant association of education with the slope of delayed recall ($y_{B\omega} = 0.023, 95\%$ CI [0.018, 0.028], p < .001), as well as a significant association of the slope of delayed recall with the slope of physical activity ($\delta_{\beta \omega} =$ 0.159, 95% CI [0.140, 0.178], p < .001). Thus, the slope of delayed recall mediated the association between education and the slope in physical activity, as confirmed by a significant indirect effect at this slope level (indirect effect S = 0.004, 95% CI [0.003, 0.005], p < 100%.001). The total effect was significant (b = 0.023, 95% CI [0.023,

N = 105,939	Primary level of education ($N = 22,526$)	Secondary level of education $(N = 59,883)$	Tertiary level of education ($N = 23,530$)	p value
Outcome: physical activity	2.3 ± 1.1	2.5 ± 0.9	2.6 ± 0.8	<.001
Mediator: cognitive function				
Delayed recall, $M \pm SD$	2.7 ± 1.8	3.8 ± 2.0	4.7 ± 2.0	<.001
Verbal fluency, $M \pm SD$	15.4 ± 5.9	20.2 ± 6.9	23.4 ± 7.4	<.001
Covariates				
Age, <i>n</i> (%)				
50-64	9,042 (40.1)	37,852 (63.2)	15,565 (66.2)	
65–79	10,949 (48.6)	19,221 (32.1)	7,038 (29.9)	
80–96	2,535 (11.3)	2,810 (4.7)	927 (3.9)	<.001
Gender		,		
Women, n (%)	13,179 (58.5)	32,369 (54.1)	11,669 (49.6)	
Men, n (%)	9,347 (41.5)	27,514 (45.9)	11,831 (50.4)	<.001

Note. Baseline = first measurement for each participant; *p* values are based on the analysis of variance and chi-square tests for continuous and categorical variables, respectively, testing the effect of education.

Table 2

Results of Longitudinal Mediation Analyses Combined With Growth Curve Models

N = 105,939	Model: delayed recall		Model: verbal fluency	
	b [95% CI]	р	b [95% CI]	р
Slope physical activity (β_{iv})				
Intercept physical activity $(\delta_{\alpha\nu})$	-0.087 [-0.093 , -0.081]	<.001	-0.080 [-0.087 , -0.074]	<.001
Education $(\gamma_{\beta\nu'})$	0.013 [0.011, 0.015]	<.001	0.010 [0.007, 0.012]	<.001
Intercept mediator $(\delta_{\alpha\omega})$	0.010 [0.008, 0.012]	<.001	0.032 [0.018, 0.037]	<.001
Slope mediator $(\delta_{\beta\alpha})$	0.159 [0.140, 0.178]	<.001	0.782 [0.719, 0.844]	<.001
Age (ref. 50–64)				
65-79	-0.021 [-0.024 , -0.018]	<.001	-0.014 [-0.017 , -0.012]	<.001
80–96	-0.067 [-0.075 , -0.060]	<.001	-0.051 [-0.059 , -0.043]	<.001
Sex (ref. women)				
Men	0.009 [0.007, 0.012]	<.001	0.006 [0.004, 0.009]	<.001
Intercept mediator $(\alpha_{i\alpha})$				
Education (v_{acc})	0.809 [0.786, 0.832]	<.001	0.365 [0.357, 0.373]	<.001
Age (ref. $50-64$)				
65–79	-0.596 [-0.629, -0.563]	<.001	-0.125 [-0.137, -0.113]	<.001
80–96	-1.458 [-1.533, -1.382]	<.001	-0.318 [-0.345 , -0.292]	<.001
Sex (ref. women)				
Men	-0.415 [-0.446, -0.384]	<.001	-0.016 [-0.027 , -0.006]	.003
Slope mediator (β_{i_0})				
Education (v_{e_0})	0.023 [0.018, 0.028]	<.001	0.002 [0.001, 0.004]	.005
Age (ref. $50-64$)				
65-79	-0.085 [-0.092 , -0.078]	<.001	-0.031 [-0.033 , -0.028]	<.001
80–96	-0.115 [-0.133, -0.097]	<.001	-0.050 [-0.056 , -0.044]	<.001
Sex (ref. women)				
Men	0.002 [-0.004 0.009]	511	-0.001 [-0.003, 0.002]	.609
Intercept physical activity (α_{m})		1011	01001 [01000, 01002]	1007
Age (ref. $50-64$)				
65-79	-0.006 [-0.023 0.012]	523	-0.001 [-0.018, 0.017]	.970
80-96	-0.304 [-0.342, -0.265]	< 001	-0.295 [-0.333 , -0.256]	< 001
Sex (ref. women)	0.2001 [0.2.12, 0.2005]	(1001	01290 [01000, 01200]	(1001
Men	0.038 [0.022, 0.054]	< 001	0.036 [0.020, 0.052]	< 001
Indirect effects	0.0000 [0.0022, 0.000 1]	(1001	01020 [01020, 01022]	(1001
Indirect effect 1 (intercept) ($v \times \delta$)	0.008 [0.007 0.010]	< 001	0.012 [0.010, 0.013]	< 001
Indirect effect 2 (slope) $(y_{\alpha\alpha} \times \delta_{\alpha})$	0.004 [0.003, 0.005]	< 001	0.002 [0.001, 0.003]	.005
Total effect			0.002 [0.001, 0.000]	.505
$\gamma_{0,i}$ ($\gamma_{0,i}$ + $\gamma_{i,i}$ × $\delta_{i,i}$ + $\gamma_{0,i}$ × $\delta_{0,i}$)	0.025 [0.023, 0.027]	< 001	0 024 [0 021 0 026]	< 001
τργ τρυ Ι σαω Α σαω Ι σβω Α σβω)	0.025 [0.025, 0.027]	1.001	3.021 [0.021, 0.020]	3.501

Note. 95% CI = confidence interval at 95%. The score of verbal fluency was divided by ten to facilitate the convergence of the models.

0.027], p < .001). Academic education remained significantly associated with change in physical activity after adjustment for delayed recall ($\gamma_{\beta y'} = 0.013$, 95% CI [0.011, 0.015], p < .001), suggesting a partial mediation. Specifically, the percentage of the total effect of education on the slope in physical activity that was mediated by the intercept and the slope of delayed recall was 48% (32% at was mediated at the intercept level, and 16% at the slope level). The model explained 33.5% of the variance in the slope in physical activity.

Verbal Fluency

We observed a significant association of education with the intercept of verbal fluency ($y_{\alpha\omega} = 0.365, 95\%$ CI [0.357, 0.373], p < .001) and a significant association of the intercept of verbal fluency with the slope of physical activity ($\delta_{\alpha\omega} = 0.032, 95\%$ CI [0.018, 0.037], p < .001). Thus, the intercept of verbal fluency mediated the association between education and the slope in physical activity, as confirmed by a significant indirect effect at this intercept level (Indirect Effect 1 = 0.012, 95\% CI [0.010, 0.013], p < .001). In addition, results showed a significant association of

education with the slope of verbal fluency ($y_{\beta\omega} = 0.002, 95\%$ CI [0.001, 0.004], p = .005), as well as a significant association of the slope of verbal fluency with the slope of physical activity ($\delta_{\beta \omega} =$ 0.782, 95% CI [0.719, 0.844], p < .001). Thus, the slope of verbal fluency mediated the association between education and the slope in physical activity, as confirmed by a significant indirect effect at this slope level (Indirect Effect 2 = 0.002, 95% CI [0.001, 0.003], p =.005). The total effect was significant (b = 0.024, 95% CI [0.021, 0.026], p < .001). Education remained significantly associated with change in physical activity after adjustment for verbal fluency $(\gamma_{\beta y'} = 0.010, 95\% \text{ CI} [0.007, 0.012], p < .001)$, suggesting a partial mediation. Specifically, the percentage of the total effect of education on the slope in physical activity that was mediated by the intercept and the slope of verbal was 58.3% (50% at was mediated at the intercept level, and 8.3% at the slope level). The model explained 43.9% of the variance in the slope in physical activity.

Sensitivity Analyses

Table 3 shows the results of the sensitivity analysis in which two additional paths were included in the model (i.e., from education to

Figure 2





 $p^* p = .05. p^{***} p < .001.$

the intercept of physical activity and from the intercept of cognitive function to the slope of cognitive function). Results were consistent with the main analyses. Specifically, the indirect effects of education on change in physical activity through both the intercept and the slope of delayed recall were significant. Education remained significantly associated with change in physical activity after adjusting for delayed recall ($\gamma_{\beta y'} = 0.010, 95\%$ CI [0.007, 0.012], p < .001), suggesting a partial mediation. Specifically, the percentage of the total effect of education on the slope of delayed recall was 68.7%. For verbal fluency, education remained significantly associated with change in physical activity after adjustment for verbal fluency ($\gamma_{\beta y'} = 0.013, 95\%$ CI [0.011, 0.015], p < .001), suggesting a partial mediation. Specifically, the percentage of the total effect of education remained significantly associated with change in physical activity after adjustment for verbal fluency ($\gamma_{\beta y'} = 0.013, 95\%$ CI [0.011, 0.015], p < .001), suggesting a partial mediation. Specifically, the percentage of the total effect of education for the slope of the total effect of education the slope of the total effect of education the slope of delayed recall was 68.7%.

on the slope in physical activity that was mediated by the intercept and the slope of verbal fluency was 53.3%.

Table 4 shows the results of the sensitivity analysis in which a time lag was introduced between cognitive function and physical activity.

The sensitivity analyses yielded results that were consistent with the main analyses. Specifically, the indirect effects of education on change in physical activity through both the intercept and the slope of delayed recall were significant. Education remained significantly associated with change in physical activity after adjusting for delayed recall ($\gamma_{\beta y'} = 0.017, 95\%$ CI [0.012, 0.022], p < .001), suggesting a partial mediation. Specifically, the percentage of the total effect of education on the slope in physical activity that was mediated by the intercept and the slope of delayed recall was 39.3%. Results of Longitudinal Mediation Analyses Combined With Growth Curve Models With Two Additional Paths Added in the Models

	Model: delayed recall		Model: verbal fluency	
N = 105,939	b [95% CI]	р	b [95% CI]	р
Slope physical activity (β_{iv})				
Intercept physical activity $(\delta_{\alpha\nu})$	-0.087 [-0.093 , -0.081]	<.001	-0.080 [-0.087 , -0.074]	<.001
Education $(\gamma_{\beta\nu'})$	0.013 [0.011, 0.015]	<.001	0.010 [0.007, 0.012]	<.001
Intercept mediator $(\delta_{\alpha\omega})$	0.010 [0.008, 0.012]	<.001	0.032 [0.028, 0.037]	<.001
Slope mediator $(\delta_{\beta\omega})$	0.159 [0.140, 0.178]	<.001	0.782 [0.719, 0.844]	<.001
Age (ref. 50–64)				
65–79	-0.021 [-0.024 , -0.018]	<.001	-0.014 [-0.017 , -0.010]	<.001
80–96	-0.067 [-0.075 , -0.060]	<.001	-0.051 [-0.059 , -0.043]	<.001
Sex (ref. women)				
Men	0.009 [0.007, 0.012]	<.001	0.006 [0.004, 0.009]	<.001
Intercept mediator $(\alpha_{i\omega})$				
Education $(y_{\alpha\alpha})$	0.809 [0.786, 0.832]	<.001	0.365 [0.357, 0.373]	<.001
Age (ref. 50–64)				
65–79	-0.596 [-0.630 , -0.563]	<.001	-0.125 [-0.137 , -0.113]	<.001
80–96	-1.458 [-1.533, -1.382]	<.001	-0.318 [-0.345, -0.292]	<.001
Sex (ref. women)				
Men	-0.415 [-0.446, -0.384]	<.001	-0.016 [-0.027 , -0.006]	.003
Slope mediator $(\beta_{i\omega})$				
Intercept mediator	-0.036 [-0.041 , -0.032]	<.001	-0.029 [-0.032 , -0.026]	<.001
Education $(y_{\beta \omega})$	0.053 [0.047, 0.059]	<.001	0.013 [0.011, 0.015]	<.001
Age (ref. 50–64)				
65–79	-0.106 [-0.113, -0.099]	<.001	-0.034 [-0.036, -0.032]	<.001
80–96	-0.168 [-0.185, -0.151]	<.001	-0.059 [-0.064 , -0.053]	<.001
Sex (ref. women)				
Men	-0.013 [-0.019, -0.007]	<.001	-0.001 [-0.003, 0.001]	.301
Intercept physical activity (α_{iv})				
Education	0.101 [0.089, 0.113]	<.001	0.105 [0.093, 0.117]	<.001
Age (ref. 50–64)				
65–79	-0.006 [-0.023, 0.012]	.524	-0.001 [-0.018, 0.017]	.972
80–96	-0.304 [-0.342, -0.265]	<.001	-0.295 [-0.333, -0.256]	<.001
Sex (ref. women)				
Men	0.038 [0.022, 0.054]	<.001	0.036 [0.020, 0.052]	<.001
Indirect effects				
Indirect effect 1 (intercept) $(y_{\alpha\omega} \times \delta_{\alpha\omega})$	0.008 [0.007, 0.010]	<.001	0.012 [0.010, 0.013]	<.001
Indirect effect 2 (slope) $(y_{\beta\omega} \times \delta_{\beta\omega})$	0.008 [0.007, 0.010]	<.001	0.010 [0.009, 0.012]	<.001
Total effects				
$\gamma_{\beta\gamma} \left(\gamma_{\beta\gamma'} + y_{\alpha\omega} \times \delta_{\alpha\omega} + y_{\beta\omega} \times \delta_{\beta\omega} \right)$	0.030 [0.027, 0.032]	<.001	0.032 [0.030, 0.034]	<.001

Note. 95% CI = confidence interval at 95%. The score of verbal fluency was divided by ten to facilitate the convergence of the models.

For verbal fluency, the result patterns were identical to the main results, although the indirect effect of education through the intercept of verbal fluency did not reach significance (p = .053) because the initial status of verbal fluency was not significantly associated with the change in physical activity ($\delta_{\alpha\omega} = 0.009, 95\%$ CI [-0.001, 0.018], p = .053). Consistent with the main analysis, education remained significantly associated with change in physical activity after adjustment for verbal fluency ($\gamma_{\beta y'} = 0.020, 95\%$ CI [0.015, 0.025], p < .001), suggesting a partial mediation. Specifically, the percentage of the total effect of education on the slope in physical activity that was mediated by the intercept and the slope of verbal fluency was 28.6%.

Discussion

Main Findings

This large-scale study of European adults aged 50 years or older revealed that the association between education level and change in physical activity behavior was explained not only by the initial status of cognitive function, but also its change across time. Higher educational level was associated with a higher initial status and a slower decline of cognitive function, which in turn slowed down the decline in physical activity. Education level remained significantly associated with physical activity after adjustment for cognitive function, suggesting that additional mechanisms may explain the relationship between academic education and physical activity. Hence, our study lends support for the long-lasting effect of education on cognitive function and, consequently, on physical activity. In this section, we discuss our results in the context of previous literature, elaborate on the mechanisms likely to explain the observed associations, and provide some limiting conditions to our findings.

Comparison With Previous Studies

Our results revealed an association between education and physical activity that is consistent with earlier evidence indicating that

Table 4

Results of Longitudinal Mediation Analyses Combined With Growth Curve Models With a Temporal Lag Between Cognitive Function and Physical Activity

	Model: delayed recall		Model: verbal fluency	
N = 59,194	<i>b</i> [95% CI]	р	b [95% CI]	p
Slope physical activity (β_{iv})				
Intercept physical activity (δ_{m})	-0.073 [-0.084 , -0.062]	<.001	-0.078 [-0.089 , -0.067]	<.001
Education $(\gamma_{\beta y'})$	0.017 [0.012, 0.022]	<.001	0.020 [0.015, 0.025]	<.001
Intercept mediator $(\delta_{\alpha\alpha})$	0.005 [0.001, 0.009]	.007	0.009 [-0.001, 0.018]	.053
Slope mediator $(\delta_{\beta\omega})$	0.172 [0.137, 0.207]	<.001	0.786 [0.639, 0.932]	<.001
Age (ref. 50–64)				
65–79	-0.032 [-0.039 , -0.025]	<.001	-0.031 [-0.038 , -0.024]	<.001
80–96	-0.064 [-0.081 , -0.047]	<.001	-0.063 [-0.081 , -0.045]	<.001
Sex (ref. women)				
Men	0.008 [0.002, 0.014]	.006	0.009 [0.003, 0.015]	.002
Intercept mediator $(\alpha_{i\alpha})$				
Education $(y_{\alpha\alpha})$	0.789 [0.757, 0.821]	<.001	0.373 [0.362, 0.384]	<.001
Age (ref. 50–64)				
65–79	-0.550 [-0.598 , -0.503]	<.001	-0.127 [-0.143 , -0.111]	<.001
80–96	-1.353 [-1.471, -1.235]	<.001	-0.308 [-0.348 , -0.268]	<.001
Sex (ref. women)				
Men	-0.438 [-0.482 , -0.393]	<.001	-0.018 [-0.032 , -0.003]	.019
Slope mediator ($\beta_{i\omega}$)				
Education $(y_{\beta \omega})$	0.040 [0.032, 0.049]	<.001	0.006 [0.003, 0.009]	<.001
Age (ref. 50–64)				
65–79	-0.101 [-0.114 , -0.088]	<.001	-0.026 [-0.030 , -0.022]	<.001
80–96	-0.146 [-0.181, -0.110]	<.001	-0.042 [-0.054 , -0.031]	<.001
Sex (ref. women)				
Men	-0.002 [-0.014 , 0.010]	.726	-0.004 [-0.007 , 0.001]	.064
Intercept physical activity (α_{iv})				
Age (ref. 50–64)				
65–79	-0.070 [-0.096 , -0.045]	<.001	-0.068 [-0.094 , -0.043]	<.001
80–96	-0.546 [-0.609 , -0.783]	<.001	-0.545 [-0.608, -0.483]	<.001
Sex (ref. women)				
Men	0.040 [0.017, 0.064]	.001	0.038 [0.014, 0.062]	.002
Indirect effects				
Indirect effect 1 (intercept) $(y_{\alpha\alpha} \times \delta_{\alpha\alpha})$	0.004 [0.001, 0.007]	.007	0.003 [-0.001, 0.007]	.053
Indirect effect 2 (slope) $(y_{\beta\omega} \times \delta_{\beta\omega})$	0.007 [0.005, 0.009]	<.001	0.005 [0.002, 0.007]	<.001
Total effects				
$\gamma_{\beta\gamma} (\gamma_{\beta\gamma'} + y_{\alpha\omega} \times \delta_{\alpha\omega} + y_{\beta\omega} \times \delta_{\beta\omega})$	0.028 [0.024, 0.032]	<.001	0.028 [0.024, 0.032]	<.001

Note. 95% CI = confidence interval at 95%. The score of verbal fluency was divided by ten to facilitate the convergence of the models.

individuals with a higher level of education are more likely to engage in a wide range of health-protective behaviors (Conti et al., 2010; Eide & Showalter, 2011; Silles, 2009; Zajacova & Lawrence, 2018), including physical activity (Chalabaev et al., 2022; Cheval, Sieber, et al., 2018; Droomers et al., 2001; Kari et al., 2020). The link between education and a sustained engagement in health behaviors can be explained by several mechanisms, including, but not limited to, increased knowledge of the health recommendations (i.e., health literacy; Friis et al., 2016; Levin-Zamir et al., 2016; Sun et al., 2013; Van Der Heide et al., 2013), a greater motivation or ability to engage in protective health behaviors (Osborn et al., 2011; Torres & Marks, 2009; Verhoeven & Snow, 2001), and better physical and social conditions (Cheval et al., 2019; Diez Roux & Mair, 2010; Rees-Punia et al., 2018; Xiao et al., 2018). Here, we demonstrate, for the first time, that this association could be underpinned by improved cognitive function. Importantly, the implementation of longitudinal mediation analyses combined with growth curve models revealed that both a higher initial status and a slower decline in cognitive function could explain the positive association between higher academic education and a slower decline in physical activity.

These findings contrast with existing literature that failed to robustly demonstrate an association between education and the rate of change in cognitive function in old age (Lövdén et al., 2020). Specifically, the previous literature suggests positive effects of educational attainment on cognitive function level, but also shows weak evidence, at best, of an association between education and cognitive decline in old age. These findings suggest that educational attainment may primarily affect late-life cognitive function through its influence on the level of cognitive function during early adulthood. Note, however, that according to the Strachan-Sheikh model of life course functioning (Vineis et al., 2016), while early life exposures (such as education) are expected to strongly impact the highest level of health functioning, they are also expected to influence aging trajectories. To explain our results suggesting an association between educational attainment and cognitive decline in old age, some features of our methods could be put forth as they increased the statistical power of our study compared to the previous literature. Specifically, Hertzog et al. (2006) argued that a lower statistical power to detect individual differences in change (vs. to detect individuals differences in the levels of cognitive function) could result from a combination of a rather limited change in cognitive function during adulthood until older age and a small number of longitudinal waves covering a relatively short time period (Brandmaier et al., 2018; Ghisletta et al., 2020). Noteworthy, our study combines a large-sample size (i.e., n > 100,000), a high number of measurements with a short time intervals (up to eight measurements with an average 2-year interval between waves), and a long follow-up (i.e., up to 15 year). These methodological features contributed to increase statistical power and our ability to detect significant associations. In addition, we used two indicators of cognitive function (i.e., delayed recall and verbal fluency), representing tow cognitive domains (Harvey, 2019), that yield consistent results. Note, however, that word-fluency tasks typically require a mixture of fluid (e.g., processing speed) and crystalized (e.g., word knowledge) cognitive abilities (Salthouse, 2005). With aging, fluid cognitive abilities typically become slower and less efficient (Baltes, 1997), but the rate at which this decline occurs greatly varies across individuals (Hartshorne & Germine, 2015). Although these features make the investigations of these abilities, and of their decline in old age, particularly relevant, the current results cannot be generalized to other cognitive abilities. In other words, it cannot be excluded that the significant association observed between education and cognitive decline could be specific to the type of tasks used in the study. In any case, since previous literature suggests a negligible effect of education on the rate of change of cognitive function in older age, our current findings should be interpreted cautiously. Additional large-scale studies with long-term follow-ups and testing multiple indicators of cognitive functioning are thus needed to replicate our findings.

Potential Mechanisms

How can this mediating effect of cognitive function on the relationship between education and physical activity be explained? Our results showing that higher educational level is associated with better cognitive function align with the previous literature (Aartsen et al., 2019; Anstey & Christensen, 2000; Avila et al., 2009; Lenehan et al., 2015; Schneeweis et al., 2014). The mechanism of cognitive reserve can be put forward to explain how education enhances cognitive function (Cullati et al., 2018; Lenehan et al., 2015; Stern, 2002, 2009). Originally, cognitive reserve was developed to explain the gap between biological damage in the brain (or brain pathology) and patients' clinical presentation (Stern, 2002). Recently, the relevance of cognitive reserve has been extended to physiological cognitive aging (Stern, 2009). Cognitive reserve is the brain's ability to optimize its performance through differential recruitment of brain networks, which may be associated with different cognitive strategies (Stern, 2002). The construction of cognitive reserve is based on cognitively stimulating situations, such as academic, occupational, and leisure activities (Opdebeeck et al., 2016; Stern, 2009). Particularly, Lenehan et al. (2015) contended that empirical data support the proposition that more educated individuals maintain a higher level of cognitive functioning compared with less educated individuals. Our findings are consistent with this proposition as they demonstrate that higher levels of education are associated with a higher initial status in cognitive function and its slower decline across time. In other words, individuals with higher levels of education may exhibit a higher cognitive reserve that is better maintained over time compared to

less educated individuals. Consequently, our results suggest that gaps in cognitive function and physical activity between the least and most educated widen over the course of aging.

We found that higher cognitive function was associated with higher physical activity. This result is consistent with previous studies suggesting that cognitive function is important to engage in physical activity (Cheval, Boisgontier, et al., 2022; Cheval, Csajbók, et al., 2021; Cheval, Orsholits, et al., 2020; Daly et al., 2015; Sabia et al., 2017). The recent TEMPA allows to account for these findings (Cheval & Boisgontier, 2021). Anchored in an evolutionary perspective, TEMPA posits that humans have an innate attraction to physical effort minimization (Cheval, Daou, et al., 2020; Cheval & Boisgontier, 2021; Cheval et al., 2017; Klein-Flügge et al., 2016; Prévost et al., 2010) and that engaging in physically active behaviors is associated with higher levels of cognitive function to overcome this attraction (Cheval, Daou, et al., 2020; Cheval, Cabral, et al., 2021; Cheval, Orsholits, et al., 2020; Cheval, Tipura, et al., 2018). This hypothesis is supported by the current results showing that education has an indirect effect on the change in physical activity across time through effects on the initial status and change in cognitive function. Although education remained significantly associated with change in physical activity after adjusting for cognitive function, these findings suggest that, among the multiple potential mechanisms that have been shown to underpin the association between education level and physical activity (i.e., health knowledge, motivation, environmental/neighborhood conditions), cognitive function may also be involved.

Limitations and Strengths

The study includes at least four limitations. First, we relied on a self-reported measure of physical activity. This measure may have reduced measurement validity and could have generated a misclassification bias (Prince et al., 2008). Likewise, this measure was used in its raw metric and treated as continuous. Yet, the transition from 1 (i.e., more than once a week) to 2 (i.e., once a week) is not equivalent to the transition from 2 to 3 (i.e., one to three times a month) or from 3 to 4 (i.e., hardly ever or never). Accelerometer-measured physical activity should be included in future studies. The same critique can be used for the education variable, which was also treated as continuous. Second, our measure of cognitive function was based on two tests intended to capture two dimensions of cognitive function, namely cognitive impairment and early signs of dementia (Harris & Dowson, 1982; Rosen, 1980; Zhao et al., 2012). Yet, general cognitive function is underpinned by many other cognitive domains such as problem solving, reasoning, processing speed, and inhibition (Davies et al., 2018; Fawns-Ritchie & Deary, 2020). Since both the effect of education and cognitive function could differ by cognitive dimension (Cheval & Boisgontier, 2021; Cheval, Cabral, et al., 2021; Cheval, Daou, et al., 2020; Cheval et al., 2019; Cheval, Radel, et al., 2018; Cheval, Tipura, et al., 2018), future studies should investigate the relationships between education, other specific cognitive functions, and physical activity. Third, our study was based on correlational data. Accordingly, we cannot exclude reverse causality and cannot infer a true causal relationship from education to physical activity through cognitive function. Yet, it should be noted that results of the sensitivity analysis, in which a time lag was introduced between the change in cognitive function and the change in physical activity to respect the temporal precedence, revealed consistent results with those observed in the main analyses. Fourth, several additional mechanisms (e.g., health literacy, motivation, environmental conditions) may explain why education is related to physical activity. Yet, these variables were not available in SHARE, which prevented us from assessing the unique association of cognitive function with physical activity after adjustment for these additional variables. As such, the proportions of the total effect of education on physical activity explained by cognitive function could have been overestimated.

The study includes at least four strengths. First, it was based on a large sample of noninstitutionalized older adults 50 years of age or older from 28 countries. Second, this is the first study to directly test and support the hypothesized mediating role of cognitive function on the association between educational attainment and physical activity. Third, we relied on a statistical approach suited to formally test mediation processes both at the baseline level and growth across time. Finally, the results of the sensitivity analyses including a time lag between the predictors and the outcome in order to minimize the risk of reverse causation bias were consistent with the main results.

Conclusion

Our findings reveal that initial status and change in cognitive function explain the positive association between educational attainment and a physical activity maintenance over the years. These results shed light on the long-lasting effect of education on the trajectories of cognitive function across aging that in turn impacts physical activity. Education, via its self-perpetuating effect in supporting cognitive function, thus plays an essential role in promoting an active lifestyle.

References

- Aartsen, M. J., Cheval, B., Sieber, S., Van der Linden, B. W., Gabriel, R., Courvoisier, D. S., Guessous, I., Burton-Jeangros, C., Blane, D., Ihle, A., Kliegel, M., & Cullati, S. (2019). Advantaged socioeconomic conditions in childhood are associated with higher cognitive functioning but stronger cognitive decline in older age. *Proceedings of the National Academy of Sciences*, 116(12), 5478–5486. https://doi.org/10.1073/pnas.1807679116
- Anstey, K., & Christensen, H. (2000). Education, activity, health, blood pressure and apolipoprotein E as predictors of cognitive change in old age: A review. *Gerontology*, 46(3), 163–177. https://doi.org/10.1159/ 000022153
- Avila, R., Moscoso, M. A. A., Ribeiz, S., Arrais, J., Jaluul, O., & Bottino, C. M. (2009). Influence of education and depressive symptoms on cognitive function in the elderly. *International Psychogeriatrics*, 21(3), 560–567. https://doi.org/10.1017/S1041610209008928
- Baltes, P. B. (1997). On the incomplete architecture of human ontogeny. Selection, optimization, and compensation as foundation of developmental theory. *American Psychologist*, 52(4), 366–380. https://doi.org/10 .1037/0003-066X.52.4.366
- Beenackers, M. A., Kamphuis, C. B., Giskes, K., Brug, J., Kunst, A. E., Burdorf, A., & van Lenthe, F. J. (2012). Socioeconomic inequalities in occupational, leisure-time, and transport related physical activity among European adults: A systematic review. *The International Journal of Behavioral Nutrition and Physical Activity*, 9(1), Article e116. https:// doi.org/10.1186/1479-5868-9-116
- Blondell, S. J., Hammersley-Mather, R., & Veerman, J. L. (2014). Does physical activity prevent cognitive decline and dementia? A systematic review and meta-analysis of longitudinal studies. *BMC Public Health*, *14*(1), Article e510. https://doi.org/10.1186/1471-2458-14-510

- Boisgontier, M. P., Orsholits, D., von Arx, M., Sieber, S., Miller, M. W., Courvoisier, D., Iversen, M. D., Cullati, S., & Cheval, B. (2020). Adverse childhood experiences, depressive symptoms, functional dependence, and physical activity: A moderated mediation model. *Journal of Physical Activity & Health*, 17(8), 790–799. https://doi.org/10.1123/jpah .2019-0133
- Börsch-Supan, A. (2022a). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 1 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w1.800
- Börsch-Supan, A. (2022b). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 2 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w2.800
- Börsch-Supan, A. (2022c). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 4 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w4.800
- Börsch-Supan, A. (2022d). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 5 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w5.800
- Börsch-Supan, A. (2022e). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 6 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w6.800
- Börsch-Supan, A. (2022f). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 7 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w7.800
- Börsch-Supan, A. (2022g). Survey of Health, Ageing and Retirement in Europe (SHARE) Wave 8 (Release version: 8.0.0) [Data set]. SHARE-ERIC. https://doi.org/10.6103/SHARE.w8.800
- Börsch-Supan, A., Brandt, M., Hunkler, C., Kneip, T., Korbmacher, J., Malter, F., Schaan, B., Stuck, S., Zuber, S., & the SHARE Central Coordination Team. (2013). Data resource profile: The Survey of Health, Ageing and Retirement in Europe (SHARE). *International Journal of Epidemiology*, 42(4), 992–1001. https://doi.org/10.1093/ije/dyt088
- Brandmaier, A. M., von Oertzen, T., Ghisletta, P., Lindenberger, U., & Hertzog, C. (2018). Precision, reliability, and effect size of slope variance in latent growth curve models: Implications for statistical power analysis. *Frontiers in Psychology*, 9, Article e294. https://doi.org/10.3389/fpsyg .2018.00294
- Chalabaev, A., Sieber, S., Sander, D., Cullati, S., Maltagliati, S., Sarrazin, P., Boisgontier, M. P., & Cheval, B. (2022). Early-life socioeconomic circumstances and physical activity in older age: Women pay the price. *Psychological Science*, 33(2), 212–223. https://doi.org/10.1177/ 09567976211036061
- Cheong, J., Mackinnon, D. P., & Khoo, S. T. (2003). Investigation of mediational processes using parallel process latent growth curve modeling. *Structural Equation Modeling*, 10(2), 238–262. https://doi.org/10 .1207/S15328007SEM1002_5
- Cheval, B. (2023). Initial status and change in cognitive function mediate the association between academic education and physical activity in adults over 50 years of age [Data set]. Zenodo. https://doi.org/10.5281/zenodo .7520036
- Cheval, B., & Boisgontier, M. P. (2021). The theory of effort minimization in physical activity. *Exercise and Sport Sciences Reviews*, 49(3), 168–178. https://doi.org/10.1249/JES.00000000000252
- Cheval, B., Boisgontier, M. P., Sieber, S., Ihle, A., Orsholits, D., Forestier, C., Sander, D., & Chalabaev, A. (2022). Cognitive functions and physical activity in aging when energy is lacking. *European Journal of Ageing*, 19(3), 533–544. https://doi.org/10.1007/s10433-021-00654-2
- Cheval, B., Cabral, D. A. R., Daou, M., Bacelar, M., Parma, J. O., Forestier, C., Orsholits, D., Maltagliati, S., Sander, D., & Boisgontier, M. P. (2021). Inhibitory control elicited by physical activity and inactivity stimuli: An EEG study. *Motivation Science*, 7(4), 386–389. https://doi.org/10.1037/ mot0000236
- Cheval, B., Csajbók, Z., Formánek, T., Sieber, S., Boisgontier, M. P., Cullati, S., & Cermakova, P. (2021). Association between physical-activity

trajectories and cognitive decline in adults 50 years of age or older. *Epidemiology and Psychiatric Sciences*, *30*, Article e79. https://doi.org/10.1017/S2045796021000688

- Cheval, B., Daou, M., Cabral, D. A. R., Bacelar, M., Parma, J. O., Forestier, C., Orsholits, D., Sander, D., Boisgontier, M. P., & Miller, M. W. (2020). Higher inhibitory control is required to escape the innate attraction to effort minimization. *Psychology of Sport and Exercise*, *51*, Article e101781. https://doi.org/10.1016/j.psychsport.2020.101781
- Cheval, B., Darrous, L., Choi, K. W., Klimentidis, Y. C., Raichlen, D. A., Alexander, G. E., Cullati, S., Kutalik, Z., & Boisgontier, M. P. (2023). Genetic insights into the causal relationship between physical activity and cognitive functioning. *Scientific Reports*, 13(1), Article 5310. https:// doi.org/10.1038/s41598-023-32150-1
- Cheval, B., Orsholits, D., Sieber, S., Courvoisier, D., Cullati, S., & Boisgontier, M. P. (2020). Relationship between decline in cognitive resources and physical activity. *Health Psychology*, 39(6), 519–528. https://doi.org/ 10.1037/hea0000857
- Cheval, B., Radel, R., Neva, J. L., Boyd, L. A., Swinnen, S. P., Sander, D., & Boisgontier, M. P. (2018). Behavioral and neural evidence of the rewarding value of exercise behaviors: A systematic review. *Sports Medicine*, 48(6), 1389–1404. https://doi.org/10.1007/s40279-018-0898-0
- Cheval, B., Rebar, A. L., Miller, M. M., Sieber, S., Orsholits, D., Baranyi, G., Courvoisier, D. C., Cullati, S., Sander, D., Chalabaev, A., & Boisgontier, M. P. (2019). Cognitive resources moderate the adverse impact of poor neighborhood conditions on physical activity. *Preventive Medicine*, *126*, Article e105741.
- Cheval, B., Sarrazin, P., Boisgontier, M. P., & Radel, R. (2017). Temptations toward behaviors minimizing energetic costs (BMEC) automatically activate physical activity goals in successful exercisers. *Psychology of Sport and Exercise*, 30, 110–117. https://doi.org/10.1016/j.psychsport .2017.02.006
- Cheval, B., Sieber, S., Guessous, I., Orsholits, D., Courvoisier, D. S., Kliegel, M., Stringhini, S., Swinnen, S. P., Burton-Jeangros, C., Cullati, S., & Boisgontier, M. P. (2018). Effect of early-and adult-life socioeconomic circumstances on physical inactivity. *Medicine and Science in Sports and Exercise*, 50(3), 476–485. https://doi.org/10.1249/MSS.000 0000000001472
- Cheval, B., Tipura, E., Burra, N., Frossard, J., Chanal, J., Orsholits, D., Radel, R., & Boisgontier, M. P. (2018). Avoiding sedentary behaviors requires more cortical resources than avoiding physical activity: An EEG study. *Neuropsychologia*, 119, 68–80. https://doi.org/10.1016/j.neuro psychologia.2018.07.029
- Conti, G., Heckman, J., & Urzua, S. (2010). The education-health gradient. *The American Economic Review*, 100(2), 234–238. https://doi.org/10 .1257/aer.100.2.234
- Csajbók, Z., Sieber, S., Cullati, S., Cermakova, P., & Cheval, B. (2022). Physical activity partly mediates the association between cognitive function and depressive symptoms. *Translational Psychiatry*, 12(1), Article e414. https://doi.org/10.1038/s41398-022-02191-7
- Cullati, S., Kliegel, M., & Widmer, E. (2018). Development of reserves over the life course and onset of vulnerability in later life. *Nature Human Behaviour*, 2(8), 551–558. https://doi.org/10.1038/s41562-018-0395-3
- Daly, M., McMinn, D., & Allan, J. L. (2015). A bidirectional relationship between physical activity and executive function in older adults. *Frontiers in Human Neuroscience*, 8, Article e1044. https://doi.org/10.3389/fnhum .2014.01044
- Davies, G., Lam, M., Harris, S. E., Trampush, J. W., Luciano, M., Hill, W. D., Hagenaars, S. P., Ritchie, S. J., Marioni, R. E., Fawns-Ritchie, C., Liewald, D. C. M., Okely, J. A., Ahola-Olli, A. V., Barnes, C. L. K., Bertram, L., Bis, J. C., Burdick, K. E., Christoforou, A., DeRosse, P., ... Deary, I. J. (2018). Study of 300,486 individuals identifies 148 independent genetic loci influencing general cognitive function. *Nature Communications*, 9(1), Article e2098. https://doi.org/10.1038/s41467-018-04362-x

- de Souto Barreto, P., Cesari, M., Andrieu, S., Vellas, B., & Rolland, Y. (2017). Physical activity and incident chronic diseases: A longitudinal observational study in 16 European countries. *American Journal of Preventive Medicine*, 52(3), 373–378. https://doi.org/10.1016/j.amepre .2016.08.028
- Diez Roux, A. V., & Mair, C. (2010). Neighborhoods and health. Annals of the New York Academy of Sciences, 1186(1), 125–145. https://doi.org/10 .1111/j.1749-6632.2009.05333.x
- Droomers, M., Schrijvers, C. T. M., & Mackenbach, J. P. (2001). Educational level and decreases in leisure time physical activity: Predictors from the longitudinal GLOBE study. *Journal of Epidemiology* and Community Health, 55(8), 562–568. https://doi.org/10.1136/jech .55.8.562
- Eide, E. R., & Showalter, M. H. (2011). Estimating the relation between health and education: What do we know and what do we need to know? *Economics of Education Review*, 30(5), 778–791. https://doi.org/10.1016/ j.econedurev.2011.03.009
- Fawns-Ritchie, C., & Deary, I. J. (2020). Reliability and validity of the UK Biobank cognitive tests. *PLOS ONE*, 15(4), Article e0231627. https:// doi.org/10.1371/journal.pone.0231627
- Friis, K., Lasgaard, M., Rowlands, G., Osborne, R. H., & Maindal, H. T. (2016). Health literacy mediates the relationship between educational attainment and health behavior: A Danish population-based study. *Journal* of Health Communication, 21(Suppl. 2), 54–60. https://doi.org/10.1080/ 10810730.2016.1201175
- Ghisletta, P., Mason, F., von Oertzen, T., Hertzog, C., Nilsson, L.-G., & Lindenberger, U. (2020). On the use of growth models to study normal cognitive aging. *International Journal of Behavioral Development*, 44(1), 88–96. https://doi.org/10.1177/0165025419851576
- Hamer, M., Muniz Terrera, G., & Demakakos, P. (2018). Physical activity and trajectories in cognitive function: English Longitudinal Study of Ageing. *Journal of Epidemiology and Community Health*, 72(6), 477– 483. https://doi.org/10.1136/jech-2017-210228
- Harris, S. J., & Dowson, J. H. (1982). Recall of a 10-word list in the assessment of dementia in the elderly. *The British Journal of Psychiatry*, 141(5), 524–527. https://doi.org/10.1192/bjp.141.5.524
- Hartshorne, J. K., & Germine, L. T. (2015). When does cognitive functioning peak? The asynchronous rise and fall of different cognitive abilities across the life span. *Psychological Science*, 26(4), 433–443. https://doi.org/10 .1177/0956797614567339
- Harvey, P. D. (2019). Domains of cognition and their assessment. *Dialogues in Clinical Neuroscience*, 21(3), 227–237. https://doi.org/10.31887/ DCNS.2019.21.3/pharvey
- Hertzog, C., Lindenberger, U., Ghisletta, P., & von Oertzen, T. (2006). On the power of multivariate latent growth curve models to detect correlated change. *Psychological Methods*, 11(3), 244–252. https://doi.org/10.1037/ 1082-989X.11.3.244
- Hillman, C. H., Erickson, K. I., & Kramer, A. F. (2008). Be smart, exercise your heart: Exercise effects on brain and cognition. *Nature Reviews Neuroscience*, 9(1), 58–65. https://doi.org/10.1038/nrn2298
- Kari, J. T., Viinikainen, J., Böckerman, P., Tammelin, T. H., Pitkänen, N., Lehtimäki, T., Pahkala, K., Hirvensalo, M., Raitakari, O. T., & Pehkonen, J. (2020). Education leads to a more physically active lifestyle: Evidence based on Mendelian randomization. *Scandinavian Journal of Medicine & Science in Sports*, 30(7), 1194–1204. https://doi.org/10 .1111/sms.13653
- Klein-Flügge, M. C., Kennerley, S. W., Friston, K., & Bestmann, S. (2016). Neural signatures of value comparison in human cingulate cortex during decisions requiring an effort-reward trade-off. *The Journal of Neuroscience*, 36(39), 10002–10015. https://doi.org/10.1523/ JNEUROSCI.0292-16.2016
- Lenehan, M. E., Summers, M. J., Saunders, N. L., Summers, J. J., & Vickers, J. C. (2015). Relationship between education and age-related cognitive

decline: A review of recent research. *Psychogeriatrics*, *15*(2), 154–162. https://doi.org/10.1111/psyg.12083

- Levin-Zamir, D., Baron-Epel, O. B., Cohen, V., & Elhayany, A. (2016). The association of health literacy with health behavior, socioeconomic indicators, and self-assessed health from a national adult survey in Israel. *Journal of Health Communication*, 21(Suppl. 2), 61–68. https://doi.org/10 .1080/10810730.2016.1207115
- Lieberman, D. E. (2015). Is exercise really medicine? An evolutionary perspective. *Current Sports Medicine Reports*, 14(4), 313–319. https:// doi.org/10.1249/JSR.00000000000168
- Littlefield, A. K., Sher, K. J., & Wood, P. K. (2010). Do changes in drinking motives mediate the relation between personality change and "maturing out" of problem drinking? *Journal of Abnormal Psychology*, *119*(1), 93– 105. https://doi.org/10.1037/a0017512
- Lövdén, M., Fratiglioni, L., Glymour, M. M., Lindenberger, U., & Tucker-Drob, E. M. (2020). Education and cognitive functioning across the life span. *Psychological Science in the Public Interest*, 21(1), 6–41. https:// doi.org/10.1177/1529100620920576
- MacKinnon, D. P. (2012). Introduction to statistical mediation analysis. Routledge. https://doi.org/10.4324/9780203809556
- Maydeu-Olivares, A. (2017). Maximum likelihood estimation of structural equation models for continuous data: Standard errors and goodness of fit. *Structural Equation Modeling*, 24(3), 383–394. https://doi.org/10.1080/ 10705511.2016.1269606
- Nettle, D. (2010). Why are there social gradients in preventative health behavior? A perspective from behavioral ecology. *PLOS ONE*, 5(10), Article e13371. https://doi.org/10.1371/journal.pone.0013371
- O'Donoghue, G., Kennedy, A., Puggina, A., Aleksovska, K., Buck, C., Burns, C., Cardon, G., Carlin, A., Ciarapica, D., Colotto, M., Condello, G., Coppinger, T., Cortis, C., D'Haese, S., De Craemer, M., Di Blasio, A., Hansen, S., Iacoviello, L., Issartel, J., ... Boccia, S. (2018). Socioeconomic determinants of physical activity across the life course: A "Determinants of Diet and Physical Activity" (DEDIPAC) umbrella literature review. *PLOS ONE*, *13*(1), Article e0190737. https://doi.org/ 10.1371/journal.pone.0190737
- Opdebeeck, C., Martyr, A., & Clare, L. (2016). Cognitive reserve and cognitive function in healthy older people: A meta-analysis. *Aging, Neuropsychology and Cognition*, 23(1), 40–60. https://doi.org/10.1080/ 13825585.2015.1041450
- Osborn, C. Y., Paasche-Orlow, M. K., Bailey, S. C., & Wolf, M. S. (2011). The mechanisms linking health literacy to behavior and health status. *American Journal of Health Behavior*, *35*(1), 118–128. https://doi.org/10 .5993/AJHB.35.1.11
- Prévost, C., Pessiglione, M., Météreau, E., Cléry-Melin, M.-L., & Dreher, J.-C. (2010). Separate valuation subsystems for delay and effort decision costs. *The Journal of Neuroscience*, *30*(42), 14080–14090. https://doi.org/ 10.1523/JNEUROSCI.2752-10.2010
- Prince, S. A., Adamo, K. B., Hamel, M. E., Hardt, J., Connor Gorber, S., & Tremblay, M. (2008). A comparison of direct versus self-report measures for assessing physical activity in adults: A systematic review. *The International Journal of Behavioral Nutrition and Physical Activity*, 5(1), Article e56. https://doi.org/10.1186/1479-5868-5-56
- Rees-Punia, E., Hathaway, E. D., & Gay, J. L. (2018). Crime, perceived safety, and physical activity: A meta-analysis. *Preventive Medicine*, 111, 307–313. https://doi.org/10.1016/j.ypmed.2017.11.017
- Rosen, W. G. (1980). Verbal fluency in aging and dementia. Journal of Clinical Neuropsychology, 2(2), 135–146. https://doi.org/10.1080/ 01688638008403788
- Rosseel, Y. (2012). Lavaan: An R package for structural equation modeling. *Journal of Statistical Software*, 48(2), 1–36. https://doi.org/10.18637/jss .v048.i02
- Sabia, S., Dugravot, A., Dartigues, J.-F., Abell, J., Elbaz, A., Kivimäki, M., & Singh-Manoux, A. (2017). Physical activity, cognitive decline, and risk

of dementia: 28 Year follow-up of Whitehall II cohort study. *The BMJ*, 357, Article ej2709. https://doi.org/10.1136/bmj.j2709

- Salthouse, T. A. (2005). Relations between cognitive abilities and measures of executive functioning. *Neuropsychology*, 19(4), 532–545. https:// doi.org/10.1037/0894-4105.19.4.532
- Sano, M., Raman, R., Emond, J., Thomas, R. G., Petersen, R., Schneider, L. S., & Aisen, P. S. (2011). Adding delayed recall to the Alzheimer Disease Assessment Scale is useful in studies of mild cognitive impairment but not Alzheimer disease. *Alzheimer Disease and Associated Disorders*, 25(2), 122–127. https://doi.org/10.1097/WAD.0b013e 3181f883b7
- Schneeweis, N., Skirbekk, V., & Winter-Ebmer, R. (2014). Does education improve cognitive performance four decades after school completion? *Demography*, 51(2), 619–643. https://doi.org/10.1007/s13524-014-0281-1
- Seltzer, M., Choi, K., & Thum, Y. M. (2003). Examining relationships between where students start and how rapidly they progress: Using new developments in growth modeling to gain insight into the distribution of achievement within schools. *Educational Evaluation* and Policy Analysis, 25(3), 263–286. https://doi.org/10.3102/ 01623737025003263
- Silles, M. A. (2009). The causal effect of education on health: Evidence from the United Kingdom. *Economics of Education Review*, 28(1), 122–128. https://doi.org/10.1016/j.econedurev.2008.02.003
- Stern, Y. (2002). What is cognitive reserve? Theory and research application of the reserve concept. *Journal of the International Neuropsychological Society*, 8(3), 448–460. https://doi.org/10.1017/ S1355617702813248
- Stern, Y. (2009). Cognitive reserve. *Neuropsychologia*, 47(10), 2015–2028. https://doi.org/10.1016/j.neuropsychologia.2009.03.004
- Stormacq, C., Van den Broucke, S., & Wosinski, J. (2019). Does health literacy mediate the relationship between socioeconomic status and health disparities? Integrative review. *Health Promotion International*, 34(5), e1–e17. https://doi.org/10.1093/heapro/day062
- Sun, X., Shi, Y., Zeng, Q., Wang, Y., Du, W., Wei, N., Xie, R., & Chang, C. (2013). Determinants of health literacy and health behavior regarding infectious respiratory diseases: A pathway model. *BMC Public Health*, *13*(1), Article e261. https://doi.org/10.1186/1471-2458-13-261
- Torres, R. Y., & Marks, R. (2009). Relationships among health literacy, knowledge about hormone therapy, self-efficacy, and decision-making among postmenopausal health. *Journal of Health Communication*, 14(1), 43–55. https://doi.org/10.1080/10810730802592247
- United Nations Educational. (2006). International standard classification of education 1997.
- Van Der Heide, I., Wang, J., Droomers, M., Spreeuwenberg, P., Rademakers, J., & Uiters, E. (2013). The relationship between health, education, and health literacy: Results from the Dutch adult literacy and life skills survey. *Journal of Health Communication*, 18(Suppl. 1), 172–184. https://doi.org/10.1080/10810730.2013.825668
- Verhoeven, L., & Snow, C. E. (2001). Literacy and motivation: Reading engagement in individuals and groups. Routledge. https://doi.org/10 .4324/9781410601735
- Vineis, P., Kelly-Irving, M., Rappaport, S., & Stringhini, S. (2016). The biological embedding of social differences in ageing trajectories. *Journal* of Epidemiology and Community Health, 70(2), 111–113. https://doi.org/ 10.1136/jech-2015-206089
- von Soest, T., & Hagtvet, K. A. (2011). Mediation analysis in a latent growth curve modeling framework. *Structural Equation Modeling*, 18(2), 289– 314. https://doi.org/10.1080/10705511.2011.557344
- Xiao, Q., Keadle, S. K., Berrigan, D., & Matthews, C. E. (2018). A prospective investigation of neighborhood socioeconomic deprivation and physical activity and sedentary behavior in older adults. *Preventive Medicine*, 111, 14–20. https://doi.org/10.1016/j.ypmed.2018.02.011

14

- Zajacova, A., & Lawrence, E. M. (2018). The relationship between education and health: Reducing disparities through a contextual approach. *Annual Review of Public Health*, *39*(1), 273–289. https://doi.org/10.1146/annure v-publhealth-031816-044628
- Zhao, Q., Lv, Y., Zhou, Y., Hong, Z., & Guo, Q. (2012). Short-term delayed recall of auditory verbal learning test is equivalent to long-term delayed recall for identifying amnestic mild cognitive impairment.

PLOS ONE, 7(12), Article e51157. https://doi.org/10.1371/journal.po ne.0051157

Received July 28, 2022 Revision received February 20, 2023

Accepted March 23, 2023